

# 

**Citation:** Er YX, Than LTL, Muslim A, Yap NJ, Tee MZ, Abdull-Majid N, et al. (2024) Infection patterns of scabies and tinea between inland and resettled indigenous Negrito communities in Peninsular Malaysia. PLoS Negl Trop Dis 18(9): e0012515. https://doi.org/10.1371/journal.pntd.0012515

Editor: Joshua Nosanchuk, Albert Einstein College of Medicine, UNITED STATES OF AMERICA

Received: February 9, 2024

Accepted: September 5, 2024

Published: September 26, 2024

**Copyright:** This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> <u>Commons CC0</u> public domain dedication.

Data Availability Statement: After carefully reviewing our IRB documents and approval documents with the Department of Orang Asli Development (JAKOA), we have decided to only share data upon request. The metadata in this study contains information on participants from very small nomadic villages and/or the last remaining villages of certain subtribes. Given the unique characteristics of these communities, including details on participants' height, weight, occupation, and income, it would be extremely difficult to sufficiently de-identify individuals. This is RESEARCH ARTICLE

# Infection patterns of scabies and tinea between inland and resettled indigenous Negrito communities in Peninsular Malaysia

Yi Xian Er<sup>1</sup>, Leslie Thian Lung Than<sup>2</sup>, Azdayanti Muslim<sup>3,4</sup>, Nan Jiun Yap<sup>1</sup>, Mian Zi Tee<sup>1</sup>, Nurmanisha Abdull-Majid<sup>1</sup>, Soo Ching Lee<sup>5</sup>, Shezryna Shahrizal<sup>3</sup>, Yvonne Ai Lian Lim<sup>5</sup>,<sup>1,6</sup>\*

 Department of Parasitology, Faculty of Medicine, Universiti Malaya, Kuala Lumpur, Malaysia,
 Department of Medical Microbiology, Universiti Putra Malaysia, UPM Serdang, Selangor Darul Ehsan, Malaysia, 3 Department of Medical Microbiology and Parasitology, Faculty of Medicine, Universiti Teknologi MARA (Sungai Buloh Campus), Sungai Buloh, Malaysia, 4 Institute for Biodiversity and Sustainable Development, Universiti Teknologi MARA, Shah Alam, Selangor, Malaysia, 5 Type 2 Immunity Section, Laboratory of Parasitic Diseases, National Institute of Allergy and Infectious Diseases, National Institute of Health, Bethesda, Maryland, United States of America, 6 Centre for Malaysian Indigenous Studies, Universiti Malaya, Kuala Lumpur, Malaysia

\* limailian@um.edu.my

# Abstract

Skin infections cause significant health burden and affect underserved communities such as the indigenous Negrito communities disproportionately. There is only one study that has addressed skin infections among the Negrito communities, which is the smallest and most isolated indigenous tribe in Peninsular Malaysia, with approximately 6.500 individuals remaining in northern and central Peninsular Malaysia. This study, which aims to update the infection patterns of scabies and tinea among the Negrito communities, recruited 361 participants from eight villages representing all six Negrito subtribes. The results revealed an overall skin infection prevalence of 35.6%, with scabies (11.7%), tinea versicolor (11.3%), and tinea imbricata (7.5%) as major infections, with no co-infection. Notably, infection rates were significantly higher in resettled villages (55.2%) compared to inland villages (24.8%). Scabies and tinea versicolor were more prevalent in resettled villages (21.2% and 23.6%, respectively) than inland villages (6.4% and 4.7%, respectively), while tinea imbricata was more common in inland villages (9.4% vs 3.9%). Furthermore, there exist predisposition of scabies among Kensiu. High prevalence of tinea imbricata was observed among the inland Bateg while prevalence of tinea versicolor was high among the resettled Bateg. Risk analysis revealed specific associations: scabies with Kensiu subtribe (P = 0.002), high income (P = 0.001) and underweight individuals (P = 0.009); tinea versicolor with Bateg subtribe (P = 0.003), resettled villagers (P < 0.001), males (P = 0.040), and overweight/obese individuals (P = 0.015); and tinea imbricata with Bateq (P = 0.011) and smokers (P = 0.004). These findings highlight a complex interplay between environment and lifestyle in skin infection prevalence. Addressing these infections requires targeted interventions, including regular medical care in inland villages and socio-economic support for resettled communities, considering the distinct predispositions in different village types.

especially true for readers familiar with the Malaysian context, who could potentially identify specific participants based on these attributes. Therefore, in order to fully comply with national rules and regulations governing research conduct in these minority communities, and out of respect for their sensitive cultural beliefs and practices, as well as to ensure the safety and privacy of the participants, sharing of the metadata must be restricted. We are committed to upholding the highest ethical standards and protecting the rights and wellbeing of these vulnerable populations throughout the research process. We stored the metadata in Universiti Malaya Research Data Repository (DOI: 10.22452/RD/BXUDKK) Any request regarding the metadata, kindly reach out to the coordinator of the Universiti Malaya Research Data Repository via researchdata@um.edu.my.

**Funding:** This research was funded by the Fundamental Research Grant Scheme from the Ministry of Higher Education, Malaysia, Grant Number: FRGS/1/2021/SKK06/UM/01/1 (FP017-2021) awarded to YALL, and an internal grant from Universiti Teknologi MARA, Malaysia, (Grant number: 600-RMC/DANA 5/3/MITRA SDG (1/ 2019)-16.) awarded to AM. The funders had no role in the study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

# Author summary

Skin infections (tinea and scabies) are significant health concerns among the indigenous Negrito communities in Peninsular Malaysia. Given that there was only one documented study among Negrito communities previously, the present study enrolled 361 participants from eight Negrito categorised into resettled or inland villages in Peninsular Malaysia to update the epidemiological status and risk factors. We found a high overall skin infection prevalence of 35.6%. Scabies, tinea versicolor, and tinea imbricata accounted for 11.7%, 11.3%, and 7.5%, respectively with no co-infection. Higher prevalence was observed among resettled villages (55.2%) compared to inland villages (24.8%). Risk factor analysis among these Negrito communities identified different predictors associated with each type of infection, generally linked to the type of village category. We highlighted that scabies and tinea versicolor were more common in resettled communities. However, rare infection like tinea imbricata, which require high treatment compliance, was more localized in inland villages. Village-specific measures including regular medical interventions in inland villages and socio-economic aid to resettled communities to address urban/ semi-urban poverty are essential, considering the existing predisposition of skin infections in different types of villages.

# Introduction

Skin infections refer to various cutaneous infection which include scabies, tinea and impetigo. These skin infections have been associated with draw-back in socio-economic development and infrastructure availability [1]. Hence, the indigenous populations of both the developed and developing countries are often the group who are deeply affected by these skin infections. For example, the prevalence of scabies, a common skin infections caused by *Sarcoptes scabiei* can be as high as 33.0% among the Australian indigenous communities [2] whereas tinea imbricata, a rare infections which is almost never found among the general population, has a prevalence as high as 61.0% among Dayak tribe in Indonesia [3], 18.7% among Negrito tribe in Malaysia [4], and up to 20.0% among the indigenous people in Papua New Guinea [5].

Authorities often underestimate the burden of skin infections due to their relatively low mortality rates [6]. However, skin infections like scabies and tinea may have significant impacts on the affected communities in terms of health, psychology, and the economy. Some of the most common skin infections among the Orang Asli is tinea [7], which is a group of skin fungal infection caused by dermatophyte such as Trichophyton and Microsporum or yeast like Malasezzia [8]. Colonization of the dermatophyte often lead to shedding of host epidermal layer and itchiness [9]. The clinical manifestation of the lesion also varies according to site, which form the basis of the disease name, include tinea capitis (body site: head, signs: balding of scalp), tinea unguium (body site: finger, signs: grey-coloration and corrosion of the fingernail) and tinea pedis (body site: feet area, signs: rash and desquamation of the area). In the case of tinea infections, the immune system's response to the fungus could trigger a secondary skin reaction called dermatophytid, characterize by erythematous rashes [10]. Another notable skin infection among these underserved communities is scabies, which is a skin infection caused by human mites known as Sarcoptes scabiei [11]. These mites burrow and reside within the host's epidermis layer, feeding on host secretions including sebum, leading to pruritus and inflammation [12], and hence results in the formation of papules vesicles in the affected areas and eczema [13]. Often the patients would scratch the affected areas, causing the formation of

impetigo in the affected areas [14]. Skin infections are often associated with negative images of inadequate level of cleanliness and poverty, leading to isolation and stigmatization by communities, which could have long term psychological effects on patients even after recovery [15].

Given the preponderance of skin infections amongst the indigenous communities, this study attempted to elucidate the status of skin infections among the underserved indigenous, namely the Orang Asli. The Orang Asli is a group of indigenous people living in Peninsular Malaysia with an approximate population size of 215,215 [16]. There are 3 main tribes, namely Senoi, Proto-Malay and Negrito and within each main tribe, there are six subtribes as shown in S1 Table. Similar to other indigenous population around the globe, Orang Asli suffers from a variety of skin infections which include dermatomycoses [7] and scabies [17] due to their traditional practices such as barefooting and pets/animal husbandry techniques, high level of poverty (76.9% of them are experiencing poverty) [18] and backwardness in development [19].

Skin infections among the Orang Asli were first reported in 1952 by Sharvil et al. [20]. In the same year, an epidemiological study was conducted by Polunin among the tribes, including the Lanoh subtribe (Negrito tribe), Semai subtribe (Senoi tribe), and Orang Seletar subtribe (Proto-Malay tribe). The prevalence of tinea was approximately 17.4% among 581 villagers from four villages, with tinea imbricata (9.1%) being the most common, followed by ringworm (2.8%) and tinea versicolor (5.5%). The distribution of tinea varied according to tribes and villages [4]. The study also highlighted that the prevalence of the skin infections among the Negrito communities were the highest (18.3%), followed by Semai (1.7–9.9%) and Proto-Malay (4.4%). In 1998, a study among 356 participants from the Proto-Malay tribe at Sungai Siput, Perak highlighted that 3.5% of the villagers had ringworm, 1.5% of them had tinea versicolor, 7% of them with impetigo and 11.9% of the villagers having scabies [17]. Both these studies used microscopy and medical observation to diagnose the skin infections. There was also a sero-prevalence survey of the scabies among the indigenous communities in 1996, which showed that 24.7% of the 312 participants were seropositive for anti-scabies antibodies [21].

The Negrito Orang Asli, the focus of our study, is the smallest tribe with an approximate total population of 6,500 (3.0% of the entire Orang Asli population) [16]. Of the six subtribes, the second largest subtribe of Negritos, Bateq (n = 2,190) are still practicing semi-nomadic lifestyles with some communities living in remote inland forest areas [22], whereas other subtribes, namely Jahai (n = 2,915) and Mandriq (n = 531) are inland agrarian. A previous study conducted in 1952 found that the prevalence of skin infections was highest among the Negrito Orang Asli tribe in Malaysia, with 18.7% of the population affected [4]. Tinea imbricata was the most common infection (18.3%), followed by tinea versicolor (6.1%) and tinea circinata (2.4%). As this data has not been updated in the past 72 years for the Negrito tribe, the present study is carried out amongst the Negrito Orang Asli tribe. With limited epidemiology studies conducted among the Orang Asli and only one study carried out among the Negrito tribe, there is a crucial need to update and fill the gap of knowledge and information regarding the current status of skin infections in these indigenous communities. Secondly, media news coverage and case reports over the past 5 years have highlighted skin infections among the Negrito communities [23–26] with no up-to-date thorough studies as evidence. Furthermore, prior study did not analyse the risk factors which were influencing these infections. Current available associated factors analysis with regards to tinea and scabies are either conducted among general population only.

Henceforth, a cross-sectional study was conducted to investigate the current status and risk factors of skin infections among indigenous Negrito communities in Peninsular Malaysia. The Negrito villages included in this study ranged from those in the inland and those in the reset-tled villages. They also comprised all the 6 subtribes within the Negrito tribe, namely Mandriq,

Bateq, Jahai, Kensiu, Kintaq, Lanoh to increase the coverage of representation of the Negrito communities. The findings of this study provided updated data on skin infections among indigenous people, which would be beneficial for public health officials in developing effective targeted and customized control programs to reduce skin infections among these underserved communities.

# Results

# Demographic data of the participants and study areas

As illustrated in Table 1, a total of 361 participants were recruited from 8 villages and each sampling site shows varying level of development status, income and main economic activity. There was a total of 5 inland villages (Villages A-E, with a total of 234 participants) and 3 resettled villages (Villages F-H, with a total of 127 participants). A majority (52.6%) of the

Table 1.	Basic demogra	aphic data of eac	h Orang Asli	village under t	he categories of i	nland and resettled villages.
I able I.	Duble delllogi	upilite dutu ol cue	in Orang mon	mage under t	ne cutegories or h	mana ana resettica images.

Village Status		Inland Villages					Resettled Villages			Total (N = 361)
V	illage	A	В	С	D	E	F	G	Н	
		(N = 40)	(N = 66)	(N = 38)	(N = 12)	(N = 78)	(N = 40)	(N = 51)	(N = 36)	
		n (%)	n (%)	n (%)	n (%)	n (%)	n (%)	n (%)	n (%)	N (%)
Gender	Female	17 (42.5)	29 (43.9)	23 (60.5)	5 (41.7)	45 (57.7)	20 (50.0)	6 (11.8)	19 (52.8)	164 (45.4)
	Male	23 (57.5)	37 (56.1)	15 (39.5)	7 (58.3)	33(42.3)	20 (50.0)	45 (88.2)	17 (47.2)	197 (54.6)
Subtribe	Bateq	32 (80.0)	66 (100.0)	38 (100.0)	12 (100.0)	0	0	42 (82.4)	0	190 (52.6)
	Jahai	0	0	0	0	0	0	0	36 (100.0)	36 (10.0)
	Kensiu	0	0	0	0	0	35 (87.5)	0	0	35 (9.7)
	Kintak	0	0	0	0	78 (100.0)	0	0	0	78 (21.6)
	Mandriq & Lanoh	8 (20.0)	0	0	0	0	5 (12.5)	9 (17.6)	0	22 (6.1)
Age	Adult	19 (47.5)	23 (34.8)	25 (65.8)	12 (100.0)	39 (50.0)	29 (72.5)	39 (76.5)	24 (66.7)	210 (58.2)
Group	Kids and teenagers	21 (52.5)	43 (65.2)	13 (34.2)	0	39 (50.0)	11 (27.5)	12 (23.5)	12 (33.3)	151 (41.8)
Body Mass Index	Normal	4 (10.0)	14 (21.2)	7 (18.4)	8 (66.7)	33 (42.3)	11 (27.5)	9 (17.6)	14 (38.9)	100 (27.7)
	Underweight	22 (55.0)	52 (78.8)	31 (81.6)	3 (25.0)	27 (34.6)	16 (40.0)	15 (29.4)	12 (33.3)	178 (49.3)
	Obese & overweight	14 (35.0)	0	0	1 (8.3)	18 (23.1)	13 (32.5)	27 (52.9)	10 (27.8)	83 (23.0)
Occupation	Agriculture	11 (27.5)	1 (1.5)	0	11 (91.7)	16 (20.5)	9 (22.5)	21 (41.2)	11 (30.6)	80 (22.2)
	Forestry	4 (10.0)	20 (30.3)	13 (34.2)	1 (8.3)	5 (6.4)	3 (7.5)	0	3 (8.3)	49 (13.6)
	Housewives	7 (17.5)	4 (6.1)	13 (34.2)	0	17 (21.8)	8 (20.0)	6 (11.8)	9 (25.0)	64 (17.7)
	Others	0	0	2 (5.3)	0	2 (206)	9 (22.5)	6 (11.8)	0	19 (5.3)
	Retiree	0	0	0	0	0	0	6 (11.8)	0	6 (1.7)
	Student	18 (45.0)	41 (62.1)	10 (26.3)	0	38 (48.7)	11 (27.5)	12 (23.5)	13 (36.1)	143 (39.6)
Water Usage	Treated	0	0	0	9(75.0)	0	6 (15.0)	24 (47.1)	0	42 (11.6)
	Untreated	40(100.0)	66(100.0)	12 (31.6)	0	78(100.0)	8 (20.0)	0	36 (100.0)	240 (66.5)
	Mix	0	0	26 (68.4)	3 (25.0)	0	26(65.0)	27 (52.9)	0	79(21.8%)
Income	<myr 800<="" td=""><td>38 (95.0)</td><td>66 (100.0)</td><td>31 (81.6)</td><td>12 (100.0)</td><td>78 (100.0)</td><td>25 (62.5)</td><td>39 (76.5)</td><td>36 (100.0)</td><td>325 (90.0)</td></myr>	38 (95.0)	66 (100.0)	31 (81.6)	12 (100.0)	78 (100.0)	25 (62.5)	39 (76.5)	36 (100.0)	325 (90.0)
	> MYR 800	2 (5.0)	0	7 (18.4)	0	0	15 (37.5)	12 (23.5)	0	36 (10.0)
Education	No formal education	16 (40.0)	37 (56.1)	22 (57.9)	12 (100.0)	43 (55.1)	13 (32.5)	15 (29.4)	5 (13.9)	163 (45.2)
	With education	24 (60.0)	29 (43.9)	16 (42.1)	0	35 (44.9)	27 (67.5)	36 (70.6)	31 (86.1)	198 (54.8)

\*Adult refer to group that are  $\geq$  20 years old whereas teen and kids were the group < 20 years old

\* Inland villages in general refer to the villages that were not part of the resettled program and are situated in their original locations, whereas resettled villages refer to the villages that participated in the relocation program.

https://doi.org/10.1371/journal.pntd.0012515.t001

participants were of Bateq subtribe, comprising of Bateq people in both resettled villages (Village G) and inland villages (Villages A-D). Kensiu, Kintak and Jahai subtribes were only found in Villages F, E and H, respectively. Participants from Mandriq and Lanoh subtribes were found in Village A, F and G.

Among the participants, 54.6% were male and 45.4% were female. The majority of the participants were either adults (>21 years old, 58.2%) or children (<9 years old, 41.8%). Slightly more than half of them (59.2%) were not working, with most being students, housewives, and retirees. An overwhelmingly majority of the participants (90.0%) belonged to households with a monthly household income of less than MYR 800, which is below the poverty income threshold (MYR 1,169) in Malaysia [27–29], a possible reason for the high number of underweight participants. In terms of educational attainment, 54.8% had formal education, while 45.2% of them never attended school.

The data also revealed that 21.8% of the study participants were using both treated and untreated water, while only 11.6% relied solely on treated water. This pattern was particularly prevalent in the more urbanized villages of Village G and Village F, where 52.9% and 65.0% of the participants, respectively, reported using both treated and untreated water in their daily lives due to urban poverty (water supply cut due to inability to pay utility bills). This information highlighted the irony that it was not necessarily inland villages that solely relied on untreated water and resettled villages relied on treated water as there were inland villages such as Village C and Village D using both treated (water from nearby tourism chalet) and untreated water and resettled village such as Village F which relied on untreated water (with support of modern water delivery system).

## Overall prevalence of scabies and tinea

Overall, 35.6% (CI = 30.5% - 40.6%) of the 361 participants had skin infections. The most common skin infections were scabies (11.6%; 95% CI = 8.6% - 15.5%), tinea versicolor (11.4%; 95% CI = 8.4% - 15.2%) and tinea imbricata (7.5%, 95% CI = 5.0% - 10.8%), as shown in Table 2. When the analysis was further stratified, it was found that the prevalence of skin infections in resettled villages was higher (55.2%) as compared to inland villages (24.8%). Specifically, both scabies and tinea versicolor were more pronounced (21.2% and 23.6%, respectively) in the resettled Negrito villages versus inland villages (6.4% and 4.7%, respectively) (P < 0.001). In contrast, tinea imbricata was found to be higher in the inland (9.4%) versus resettled villages (3.9%) (P = 0.049). It was interesting to note that there was no co-infection of tinea and scabies among the Negrito Orang Asli.

As illustrated in Table 2, huge variations in terms of scabies and tinea distributions were demonstrated among the villages. More than 70% of the participants in villages C, D and F had scabies and tinea, whereas village A had the lowest prevalence (7.7%). The Kensiu subtribe appeared to have the highest prevalence (60.0%) of skin infections. This Kensiu subtribe community resides in Village F, a resettled village with basic amenities such as treated water and electricity, as well as walking distance to health and education (primary and secondary) facilities.

It was also observed that Bateq subtribe living in different villages displayed predisposition toward different skin infections. For example, those Bateq living in resettled Village G displayed high prevalence of tinea versicolor (47.06%) whereas Bateq living in inland Villages C (42.11%) and D (33.33%) displayed high prevalence of tinea imbricata. There was a higher prevalence of scabies and tinea versicolor among the obese and overweight group (prevalence of 13.3% in scabies and 24.1% in tinea versicolor) and underweight participants (prevalence of 14.6% in scabies and 9.0% in tinea versicolor) than those with normal BMI (prevalence of 5.0% in both scabies and tinea versicolor).

				Overall ski n (	n infections (%)	Type of Skin Infection n (%)			ı (%)
	Variables	Description	Total Number/ N	Positive	Negative	Scabies	Tinea versicolor	Tinea imbricata	Other Tinea & skin diseases
Village	Inland Villages	Α	40	8 (20.00)	32 (80.00)	3 (7.50)	4 (10.00)	0	1 (2.50)
		В	66	6 (9.09)	60 (90.91)	2 (3.03)	2 (3.03)	2 (3.03)	0
		С	38	28 (73.68)	10 (26.32)	3 (7.89)	0	16 (42.11)	9 (23.68)
		D	12	10 (83.33)	2 (16.67)	0 (0.00)	5 (41.67)	4 (33.33)	1 (8.33)
		Ε	78	6 (7.89)	71 (92.11)	7 (8.97)	0	0	0
	Resettled	F	40	34 (85.00)	6 (15.00)	24 (60.00)	0	5 (12.50)	5 (12.50)
	Villages	G	51	30 (58.82)	21 (41.18)	3 (5.88)	24 (47.06)	0	3 (8.33)
		Н	36	6 (16.67)	30 (92.11)	0	6 (16.67)	0	0
Subtribe		Bateq	190	73 (38.42)	117 (61.58)	8 (4.21)	30 (15.79)	22 (11.58)	13 (6.84)
		Jahai	36	6 (16.67)	30 (83.33)	0	6 (16.67)	0	0
		Kensiu	35	29 (82.86)	6 (17.14)	19 (54.29)	0	5 (14.29)	5 (14.29)
		Kintak	78	7 (8.97)	71 (91.03)	7 (8.97)	0	0	0
		Mandriq & Lanoh	22	14 (63.64)	8 (36.36)	8 (36.36)	5 (22.72)	0	1 (4.55)
1	Age group	Adult	210	85 (40.48)	125 (59.52)	22 (10.48)	28 (13.33)	19 (9.05)	16 (7.62)
		Kids and teenagers	151	44 (29.14)	107 (70.86)	20 (13.24)	13 (8.61)	8 (5.30)	3 (1.99)
Gender		Male	197	80 (40.61)	117 (59.39)	23 (11.68)	32 (16.24)	12 (6.09)	13 (6.60)
		Female	164	49 (29.89)	115 (70.11)	19 (11.56)	9 (5.49)	15 (9.15)	6 (3.66)
	Total		361	128 (35.46)	233 (64.54)	42 (11.63)	41 (11.36)	27 (7.48)	19 (5.26)
95% CI (Based on total)			30.57- 40.66	59.33- 69.43	8.60- 15.51	8.36-15.20	5.08-10.82	3.29-8.24	

Table 2. Prevalence and 95% confidence interval of skin infections among the participants, in accordance with the different villages. Scabies and tinea versicolor were more prevalent among the resettled villages, whereas tinea imbricata was mostly found in inland villages.

\*Tinea and others: tinea refer to those who are positive in potassium hydroxide test but negative in CLSI-54a test whereas others refer to those with symptoms but negative in both tests

\* Inland villages in general refer to the villages that were not part of the resettled program and are situated in their original locations, whereas resettled villages refer to the villages that participated in the relocation program.

https://doi.org/10.1371/journal.pntd.0012515.t002

# Prevalence and risk factor analysis of scabies, tinea versicolor and tinea imbricata

**a Scabies.** In this study, scabies accounted for 11.6% of the prevalence rate, with 60.0% of cases belonging to the Kensiu subtribe who reside in the more urbanized village, Village F. On the other hand, its prevalence was fairly consistent across all other villages, remaining at around 3–7%. Interestingly, Villages D (Inland) and H (resettled) were free from scabies infections. There were 54.3% of the Kensiu participants and 36.4% of the Mendriq/Lanoh participants who had scabies, while the Bateq participants had the lowest prevalence of scabies (4.2%).

In the univariate analysis (Table 3), we found that individuals from the Kensiu subtribe (COR = 18.9, P < 0.001), underweight (COR = 3.3, P = 0.02), an income level less than RM 800 (COR = 7.9, P < 0.001), and those living in resettled villages had higher odds of scabies infections compared to their counterparts (COR = 3.9, P < 0.001). However, in the multivariate analysis, only three factors remained significant predictors of scabies infections: Kensiu subtribe (AOR = 7.7, P = 0.002), underweight (AOR = 9.7, P = 0.009) and income of more than RM 800 (AOR = 7.5, P = 0.001).

Variable	Ν	Scabies infected, n (%)	Univariate		Multivariate	
			COR (CI)	P value	AOR (CI)	P value
Kintak subtribe					Nil	
No	283	35 (12.4)	1	0.41		
Yes	78	7 (9.0)	0.7 (0.3–1.6)			
Kensiu subtribe						
No	324	21 (6.5)	1		1	
Yes	37	21 (56.8)	18.9 (8.7–42.4)	<0.001*	7.7(2.3–29.3)	0.002*
Bateq subtribe						
No	168	31 (18.5)	1		1	0.07
Yes	193	11 (5.7)	0.3 (0.1–0.5)	<0.001*	0.4(0.1-1.0)	
Gender					Nil	
female	164	19 (11.6)	1	0.979		
male	197	23 (11.7)	1.0 (0.5–1.9)			
Body Mass Index						
Normal	100	5 (5.0)	1		1	
Obese & overweight	83	11 (13.3)	2.9 (1.0-9.5)	0.058	3.2 (0.8-14.1)	0.109
Underweight	178	26 (14.6)	3.3 (1.3-9.9)	0.020*	9.7 (3.0-41.1)	0.009*
Age group					Nil	
Adult	210	22 (10.5)	1	0.419		
Kids and teens	151	20 (13.2)	1.3 (0.7–2.5)			
Income level						
<rm800< td=""><td>325</td><td>27 (8.3)</td><td>1</td><td>&lt;0.001*</td><td>1</td><td></td></rm800<>	325	27 (8.3)	1	<0.001*	1	
>RM800	36	15 (41.7)	7.9 (3.6–17.1)		7.5(2.3-25.0)	0.001*
Presence of pets					Nil	
No	144	15 (10.4)	1	0.557		
Yes	217	27 (12.4)	1.2 (0.6–2.4)			
Presence of family member with the same disease					Nil	
No	200	26 (13.0)	1	0.368		
Yes	161	16 (9.9)	0.7 (0.4–1.4)			
Usage of topical ointments					Nil	
No	244	18 (7.4)	1	<0.001*		
Yes	117	24 (20.5)	3.2 (1.7-6.3)			
Village status						
Inland village	234	15 (6.4)	1	<0.001*	1	0.896
Resettled village	127	27 (21.3)	3.9 (2.0-7.9)		1.1 (0.3–3.2)	
Occupations					Nil	
Away from village	148	12 (8.1)	1	0.085		
Within village	213	30 (14.1)	1.9 (0.9–3.9)			
Tobacco					Nil	
No	282	33 (11.7)	1	0.94		
Yes	79	9 (11.4)	1 (0.4–2.1)			

#### Table 3. Potential risk factors associated with scabies based on logistic regression analysis.

Number in dataframe = 361, Number in model = 361, Missing = 0, AIC = 197.5, C-statistic = 0.82, H&L = Chi-sq(8) 4.96 (P = 0.762)

https://doi.org/10.1371/journal.pntd.0012515.t003

# b Tinea versicolor

There were 11.4% of the participants infected with tinea versicolor. The prevalence in resettled villages (Village F, G, and H) was at 23.6% which was much higher than in inland villages

Variable	N	Tinea versicolor infected, n (%)	Univariate		Multivariate	
			COR (CI)	P value	AOR (CI)	P value
Jahai subtribe					Nil	
No	325	35 (10.8)	1	0.294		
Yes	36	6 (16.7)	1.7 (0.6-4.0)			
Bateq subtribe						
No	168	11 (6.5)	1	0.009*	1	0.001*
Yes	193	30 (15.5)	2.6 (1.3-5.7)		4.4 (1.9-10.9)	
Gender						
female	164	9 (5.5)	1	0.002*	1	0.040*
male	197	32 (16.2)	3.3 (1.6-7.6)		2.6 (1.1-6.7)	
Body Mass Index						
Normal	100	5 (5.0)	1		1	
Obese & overweight	83	20 (24.1)	6.0 (2.3–18.9)	0.001*	4.3 (1.4–15.1)	0.015*
Underweight	178	16 (9.0)	1.9 (0.7–5.9)	0.234	2.9 (0.9–10.4)	0.081
Age group					Nil	
Adult	210	28 (13.3)	1	0.166		
Kids and teens	151	13 (8.6)	0.6 (0.3-1.2)			
Presence of pets					Nil	
No	144	21 (14.6)	1	0.118		
Yes	217	20 (9.2)	0.6 (0.3-1.1)			
Presence of family member with the same disease						
No	200	14 (7.0)	1	0.005*	1	0.003*
Yes	161	27 (16.8)	2.7 (1.4-5.4)		3.5 (1.6-8.2)	
Usage of topical ointments					Nil	
No	244	32 (13.1)	1	0.133		
Yes	117	9 (7.7)	0.6 (0.2–1.2)			
Occupations					Nil	
Away from village	148	18 (12.2)	1	0.688		
Within village	213	23 (10.8)	0.9 (0.4–1.7)			
Village status						
Inland village	234	11 (4.7)	1	<0.001*	1	<0.001*
Resettled village	127	30 (23.6)	6.3 (3.1–13.6)		8.9 (3.7-23.8)	
Tobacco						
No	282	27 (9.6)	1	0.047*	1	0.23
Yes	79	14 (17.7)	2.0 (1.0-4.1)		1.7 (0.7–3.8)	

#### Table 4. Potential risk factors associated with tinea versicolor based on logistic regression analysis.

Number in dataframe = 361, Number in model = 361, Missing = 0, AIC = 203.9, C-statistic = 0.834, H&L = Chi-sq(8) 11.12 (P = 0.195)

https://doi.org/10.1371/journal.pntd.0012515.t004

(11.6%). Similarly, Bateq (15.5%) had a higher prevalence than non-Bateq participants (6.5%). Tinea versicolor was particularly pronounced in Village G (47.06%), a resettled Bateq villages. Furthermore, males displayed a higher prevalence (16.2%) compared to females (5.5%), and the overweight and obese group (24.1%) had a higher prevalence than underweight (9.0%) and normal (5.0%) groups (Table 4). Interestingly, 16.8% of the participants had family member(s) in the same household with tinea versicolor, which is more than double those who did not have any (7.0%). Pet owners (9.2%) had a lower prevalence than non-pet owners (14.2%).

In the univariate analysis (Table 4), Bateq subtribe (COR = 2.6, P = 0.009), males (COR = 3.3, P = 0.002), obese and overweight individuals (COR = 6.0, P = 0.001), presence of

family members with tinea versicolor (COR = 2.7, P = 0.005), living in resettled villages (COR = 6.3, P < 0.001), and smokers (COR = 2.0, P = 0.047) had higher risks of getting tinea versicolor. Multivariate analysis showed Bateq subtribes (AOR = 4.4, P = 0.001), resettled villagers (AOR = 8.9, P < 0.001), males (AOR = 2.6, P = 0.040), obese and overweight individuals (AOR = 4.3, P = 0.015) and presence of family members with tinea versicolor (AOR = 3.5, P = 0.003) had a higher risk of getting tinea versicolor.

#### c Tinea imbricata

There were 7.5% participants who had tinea imbricate. More than half of those infected belonged to the inland Bateq subtribe in Villages C and D, underweight and not receiving any formal education. Chronic infections seemed to be a common trend, as 74.0% of them were infected with tinea imbricata for >100 weeks (~2 years) (S5 Table). In contrast with scabies and tinea versicolor, tinea imbricate had higher prevalence in inland villages (9.4%) compared to resettled villages (3.9%). Furthermore, female (9.1%) had higher prevalence than male (6.1%) (Table 5).

Both univariate and multivariate analyses (Table 5) highlighted that individuals from the Bateq subtribe (COR = 4.2, P = 0.005, AOR = 3.7, P = 0.011, respectively) and smokers (COR = 3.8, P = 0.001, AOR = 3.3, P = 0.004, respectively) had higher odds of getting tinea imbricata.

# Discussion

Negrito communities are unique due to their distinct lifestyles, which differ from other indigenous tribes. These communities exhibit a broad spectrum of living conditions, ranging from isolated and primitive inland villages to resettled peripheral urban areas [30]. This diversity makes them well-suited for studying the interplay between urbanization and traditional lifestyles with regards to the prevalence of skin infections. There is a growing need for additional data to accurately assess the burden and impact of skin infections among the Negrito Orang Asli.

The present results revealed that skin infections remain a significant health concern within the Negrito communities in Peninsular Malaysia. More than 35.0% of the overall participants were found to have skin infections, with prevalence exceeding 50.0% in inland villages C, D, F and G. Despite belonging to the same Negrito tribe, villages A, B, E and H have significantly lower prevalence of skin infections compared to the other villages. The low prevalence of village A and B were perhaps associated with high frequency of medical interventions, which were effective in regulating the skin infections in those areas [31]. There was also no co-infection between scabies and tinea found in our study, which is consistent with other epidemiological studies showing that scabies is associated with pyoderma [32] and impetigo [14], rather than tinea. Co-infections of scabies and tinea appear to be associated to severely immunocompromised individuals [33].

It is interesting to note that the prevalence of skin infections in resettled villages was higher (55.2%) compared to inland villages (24.8%). Similar trend was observed when it comes to other tropical diseases such as helminthiasis [34]. Although it is generally hypothesized that better economic and infrastructure provision are negatively correlated with prevalence of infections as resettled villages typically result in provision of amenities, better accessibilities, infrastructure compared to the inland villages as per outlined in Lin et al., 1994 [35], the current findings suggest that several factors may counter this trend. Resettlement often involves merging of multiple Orang Asli communities into a single, larger village, leading to higher population density [35] which can facilitate the transmission of skin diseases, especially

Variable	N	Tinea imbricata infected, n (%)	Univariate		Multivariate	
			COR (CI)	P value	AOR (CI)	P value
Bateq subtribe						
No	325	5 (3.0)	1	0.005*	1	0.011*
Yes	36	22 (11.4)	4.2 (1.7-12.8)		3.7 (1.4–11.3)	
Gender					Nil	
female	164	15 (9.1)	1	0.275		
male	197	12 (6.1)	0.6 (0.3–1.4)			
Underweight					Nil	
No	183	9 (4.9)	1	0.066		
Yes	178	18 (10.1)	2.2 (1.0-5.2)			
Income level					Nil	
<rm800< td=""><td>325</td><td>23 (7.1)</td><td>1</td><td>0.387</td><td></td><td></td></rm800<>	325	23 (7.1)	1	0.387		
>RM800	36	4 (11.1)	1.6 (0.5-4.6)			
Age group					Nil	
Adult	210	19 (9.0)	1	0.187		
Kids and teens	151	8 (5.3)	0.6 (0.2–1.3)			
Presence of family member with the same disease					Nil	
No	200	10 (5.0)	1	0.051		
Yes	161	17 (10.6)	2.2 (1.0-5.2)			
Usage of topical ointments					Nil	
No	244	23 (9.4)	1	0.052		
Yes	117	4 (3.4)	0.3 (0.1-0.9)			
Occupations					Nil	
Away from village	148	15 (10.1)	1	0.115		
Within village	213	12 (5.6)	0.5 (0.2–1.2)			
Village status					Nil	
Inland village	234	22 (9.4)	1	0.068		
Resettled village	127	5 (3.9)	0.4 (0.1–1.0)			
Tobacco						
No	282	14 (5.0)	1	0.001*	1	0.004*
Yes	79	13 (16.5)	3.8 (1.7-8.5)		3.3 (1.4–7.4)	

#### Table 5. Potential risk factors associated with tinea imbricata based on logistic regression analysis.

Number in dataframe = 361, Number in model = 361, Missing = 0, AIC = 180.2, C-statistic = 0.713, H&L = Chi-sq(8) 2.10 (P = 0.978)

https://doi.org/10.1371/journal.pntd.0012515.t005

scabies. Moreover, there are varying levels of development in the resettlement scheme depending on the economic development of the country, regular provision by the authorities, and compliance by the villages [35]. Exposure to semi-urbanization or urbanization has also caused the younger generation of Negrito Orang Asli to be uninterested in agriculture, their traditional activity due to exposure to urban lifestyles. At the same time, some may also not be interested to work in the semi-urban or urban areas as they are unable to adapt to urban lifestyles due to inadequate education attainment [36], which lead to urban poverty issues and inability to pay for water and electricity bills or afford nutritious diets although there is provision of these basic amenities.

Furthermore, there seems be an association between subtribe and disease predisposition, suggesting the complex interplay of lifestyle and living habits influence on the likelihood of skin infections. For example, Kensiu subtribe displayed strong predisposition toward scabies as all of them resided in the village F (the sole Kensiu village in Malaysia). The Kensiu people

live in the resettled village and they are more prone to higher inflation rates due to closer proximity to the urban areas resulting in urban poverty [37,38]. This was further supported by news coverage from 2020 indicating that the villagers could not sustain their livelihood in their resettled area and were forced to return to the inland forest for sustenance [39]. Moreover, unlike those in inland areas, these villagers could not rely on forest resources as part of their diet and for non-regular side income (eco-tourism, forest resource scavenging) [35]. This phenomenon leads to the inability to purchase sufficient foods, resulting in malnutrition [40] (40.0% of the participants were underweight and 32.5% were overweigh/obese, highlighting the phenomenon of double-burden malnutrition [41]). In addition, the Kensiu village is densely populated areas with abundant vegetation and shade, contributing to reduced sunlight exposure. It has been reported that factors which may facilitate the transmission of mites include increased contact among villagers and limited sunlight exposure [42].

On the other hand, different Bateq subtribes displayed different infection patterns. The Bateq in the inland villages, Village C (42.11%) and Village D (33.33%) had high prevalence of tinea imbricata whereas Bateq in resettled Village G (47.06%) displayed high prevalence of tinea versicolor. The higher prevalence of tinea imbricata among the inland villages, could be attributed to the smaller population size and gene pool, which might lead to the higher prevalence of susceptible genes due to inbreeding [43] as well as the higher exposure to the spores which remains in the areas [44]. However, the low prevalence of tinea imbricata in inland villages A and B was attributed to media coverage of tinea outbreaks in 2018 and 2019 [23,25] which prompted active intervention by health authorities and the non-governmental organizations (NGOs). Health care officers visited these villages frequently, providing villagers with medications and consistent monitoring to ensure good treatment compliance and follow-up [45,46]. It is important to note that, tinea imbricata is curable with anti-fungal such as terbina-fine [47]. The key to controlling these infections lies in follow-up to ensure patient compliance. Medical intervention and follow-up are effective measures in controlling this infection [31].

On the other hand, the high prevalence of tinea versicolor among resettled Bateq village could be attributed to the village surrounding environment and housing design (Fig 1A and 1B). In this village, there was less vegetation, resulting in limited shaded areas and high exposure to sunlight and higher temperature [48]. The high temperature which promote perspiration could induce higher risk of tinea versicolor [49], particularly in the resettled villages where the villagers were staying in brick houses (Fig 1B) which lack ventilation [50].

Scabies was the most common skin infection (11.6%) among the Negritos in this study and there have been no prior publications about its prevalence among this community. However, the prevalence found in this study is similar to that reported among the Temiar Orang Asli communities in 1998 in Pos Piah, Perak (11.9%) [17] and in suburban communities in Jengka Triangle, Pahang (11.6%) [51]. Comparatively, its prevalence is only slightly higher compared to the prevalence reported in an urban population survey in Kuching, East Malaysia (8.1%) [52] indicating that scabies are not limited to lower-income groups or less-developed areas. Further analysis indicated scabies was most prevalent (60.0%) in village F, supporting the latter statement despite the village's closer proximity to urban areas and access to basic infrastructure. Most of the cases were found among individuals categorized as underweight (14.6%), which correlate with the global research, that malnourish [53] was the key risk factors of scabies.

On the other hand, the prevalence of tinea versicolor was 11.4%, primarily observed in villages D and G, which is higher than the earlier study among the Negrito communities of Perak in 1952 (6.1%) [4] and among Senoi tribe in Pos Piah (1.7%) [17]. It is interesting to note that the prevalence of the current study is lower than the finding among the urban population in Singapore (25.2%) [54]. Globally, poverty [55], hygiene practices and sanitary infrastructure



Fig 1. Village architecture of Bateq people. (A) Bateq inland village which has limited tall vegetation and shaded areas leading to high sunlight exposure. (B) In this resettled village, villagers stay in government-built brick houses.

[56], treatment compliance [57] and malnutrition [58] and genetic [43] have been associated with the prevalence of tinea. Similar to the study in Brazil and China [59,60], males had higher odds of getting tinea versicolor. This is likely due to males having higher testosterone levels, which results in more active sweat glands [61], which promote the pathogenesis of tinea versicolor [58]. Moreover, obese and overweight group had higher odds of getting tinea versicolor due to a lower tolerance of heat and hence lead to more active sweat glands and perspiration rates [62]. Furthermore, the presence of a family member with the same infection was identified as a risk factor for tinea versicolor. This can be attributed to to complex polyclonal inheritance nature of tinea versicolor [59], meaning there are multiple genetic factors involved. The presence of a family member with tinea versicolor indicates the presence of susceptibility gene sets, which increases the likelihood of developing the condition.

Finally, the prevalence of tinea imbricata in our study (7.5%) was much lower than the report in 1952 among the Negrito communities in Perak (18.3%) [4]. The present prevalence data among Negrito is also lower compared to the observations among the Semai tribe (12.5%) in 1952, but higher compared to the Proto-Malay tribe (4.4% in 1952) [4] and among urban population (3.5% in 2002) [63]. Our multivariate binomial logistic regression model showed that smoking or the use of tobacco is an important risk factor for tinea imbricata. In this context, smoking leads to immunomodulation of the host's immunity through the generation of platelet-activating factor agonists [64] and physical changes to the skin, including lowering epidermis and dermis thickness [65]. These factors may increase the susceptibility of smokers to tinea infections compared to non-smokers.

### Limitation and future studies

Cross-sectional study such as this has inherent limitations particularly in its ability to provide only a snapshot in time, limiting the capture of dynamic changes in disease patterns and hindering the establishment of causal relationships. A longitudinal study would be essential to obtain a more comprehensive understanding of the temporal dynamics of skin infections within the indigenous population. Moreover, there is variation of the village sizes, especially the inland or nomadic to semi-nomadic villages are much smaller than those resettled villages due to mobility and logistic issues, therefore posed significant challenges in standardizing sample sizes. Future research should focus on intrinsic and biotic factors including the microbiome, hormones, cytokines, and host genetics to gain comprehensive insights into the causes and persistence of skin infections in the Negrito communities, enabling the development of tailored preventive and therapeutic strategies.

# Conclusion

This study highlighted the current burden of skin infections among the Negritos revealing a higher prevalence than previously reported. Our results displayed a strong predisposition of different infections toward different village settings: scabies and tinea versicolor are more prevalent in resettled and more advanced villages, whereas tinea imbricata is primarily found in the inland villages. Other than that, villages with the same subtribe had different predispositions due to variation in the level of development and frequency of medical interventions. Other contributing factors include lifestyle practices, such as smoking, can weaken host immunity and elevate the risk of skin infections. Variations in village lifestyles, housing conditions, population density, and sunlight exposure also influence disease risk. To combat these infections effectively, tailored approaches are needed for different village types. Inland villages may benefit from scheduled medical interventions, while resettled communities could require socio-economic support to alleviate urban or semi-urban poverty. These strategies acknowledge the varying susceptibility to skin infections observed across different village settings.

# Material and methods

# a Ethics application

This study was approved by the Medical Ethics Committee of the University of Malaya Medical Centre (UMMC) (Reference No.: 20201030–9172) to collect skin samples, photos, and questionnaires from participants and permission was also obtained from the Department of Orang Asli Development (JAKOA) [Reference No.: JAKOA.PP.30.032 JLD 52 (34)]. Additional consultations were obtained from the JAKOA official at the district level and villages' chieftains (referred to as 'Tok Batin') prior to commencing the research in the respective areas and villages. The purpose and procedures of the study were verbally explained to the villagers and printed form (both in Bahasa Melayu and English) were also provided before the commencement of the research. The interview and consent approval were conducted in Bahasa Melayu, the national language of Malaysia. If participants were unable to comprehend Bahasa Melayu, a village interpreter will be involved in translating. Prior to inclusion in this study, written consent was obtained from all participants. For children under 18 years old, an assent was obtained followed by written consent from their parents or guardians.

# b Study area and study population

This cross-sectional study was conducted between November 2021 and March 2023 in eight Negrito villages located in the Northern and Central Peninsular Malaysia (Fig 2). The focus on



Fig 2. Location of studies sites. Different villages showed different levels of development. (plotted with ArcMap v10.7, sources: Esri, Michael Bauer Research GMBH 2022, Department of Statistics Malaysia [67].

Negrito tribe was based on the study by Polunin in 1952, which indicated that Negrito tribe was the most prone to skin infections. The approval to conduct study at the villages was based on the discussions with the central and district/local JAKOA officers. Additionally, the opinions and receptiveness of the village committee and village chieftain were also taken into consideration. The villages identified were of different levels of economic activity, infrastructure availability, and accessibility to assess the impact of development on the prevalence of skin infections (Table 6). Native tongue was the primary language of the villagers though they had a good, or at least satisfactory, command of the Bahasa Melayu.

There are 2 main categories of villages, the resettled village that have been relocated under the government's resettlement plan scheme, with the new location near the township, with

Table 6.	Characteristics of each	ı village.
----------	-------------------------	------------

Village	Subtribe	Number of participants	Water sources	Accessibility	Distance from town/ hr	Main economic	Village status
А	Bateq	40	River	Tar road	2	Agriculture	Inland
В	Bateq	66	River	Tar road and mud road	2	Forestry	Inland
С	Bateq	38	River	Boat only	2.5	Forestry	Inland
D	Bateq	12	River	Tar road	2.5	Forestry	Inland
Е	Kintak	78	River	Tar road	1	Agriculture	Inland
F	Kensiu	40	River and treated water	Tar road	0.16	Industry, tourism, Agriculture	Resettled
G	Bateq	51	River and treated water	Tar road	0.6	Agriculture	Resettled
Н	Jahai	36	River	Tar road	0.6	Agriculture	Ressetled

\*Resettled village refer to those where the entire villages was moved under the Resttlement plan scheme (RPS) (34)

https://doi.org/10.1371/journal.pntd.0012515.t006

housing, water and electricity infrastructure provided [66] and those which maintained their traditional lifestyle which may include being semi-nomadic in the inland/remote villages. The resettled villages are those villages which participated in the relocation scheme collectively called *as Rancangan Pengumpulan Semula* (RPS) since 1970s whereas inland village refer to those villages which are still situated in their original locations. In brief, resettled villages were relocated by the Department of Orang Asli Development (JAKOA) from interior to a more accessible location together with promises of multiple development scheme including basic infrastructure and facility, economic subsidy, economic diversification programme (eg., ALDP [Agriculture Land Development Programme]) and eco-tourism [35].

## c Sampling location

Sampling sites were determined through multi-stakeholder discussions as described previously. Upon institutional IRB approval, discussion sessions were held with the JAKOA officers to explain the rationale and aims of the study. The JAKOA officers determined the suitability of the villages, for example, whether the village chieftain, committee and villagers are receptive to the research activities, accessibility and remoteness of the villages (certain villages can only be reached by helicopter), fluency of national language as the research is conducted in the national language.

Prior to the study being conducted in the villages, JAKOA officers accompanied researchers to the villages for meetings with the Tok Batin (the chieftain of the village) to further explain the rationale and aims of the study. Following that, the whole village was invited for a briefing and dialogue session and samples were only taken from individuals who gave consent after understanding the nature of the research.

Therefore, engagement prior to sampling was according to the national rules and regulations of research conduct in these minority communities, adhering and respecting sensitive cultural beliefs and practices.

## d Sample collection

Power calculation developed by Cohen, 1988 [68] were employed to calculate our sample sizes via R package PWR [69]. By including desired power of 90% and significant level to 0.01, the script would be as the next paragraph, with the minimum sample sizes of 231 for scabies study and 156 for tinea study (Table 7).

pwr.2p.test(h = ES.h(p1 = prevalence rate 1, p2 = prevalence rate 2), sig.level = 0.01, power = .90, alternative = "greater")

Individuals with a minimum age of 3 years old, who had resided in the village for at least a year were included in our study. The children below 3 years old were excluded from our study due to potential challenges in obtaining their assent and also the concerns about the use of glass slides during sample collection. Given that children below 3 years old might find the process distressing, it could negatively impact the participation rates in the study. Besides, non-

Disease	Prevalence (%)	Number of sample (n)	Sample sizes/n	Effect sizes/h
Scabies	24.7% (21)	312	231	0.34
	11.9% (17)	356		
Tinea	17.4% (4)	581	156	0.41
	5% (17)	356		

Table 7. Reference and output for sample size calculation.

https://doi.org/10.1371/journal.pntd.0012515.t007



Fig 3. Photo of skin infections. (A) tinea imbricata (B) tinea versicolor and (C) scabies (Photo taken by Yi Xian, Er after getting the consent from the participants).

Negrito individuals who resided in the village or Negrito individuals who were only staying in the village temporarily were also excluded from the study.

Skin scraping samples were collected from the participants following the description by Ponka (2014) [70]. Brand-new glass slides were used to gently scrape the skin and the samples were collected onto black sugar paper. The samples were then placed in pre-labelled envelopes with the participant's name and identity code. All the skin samples were preserved at ambient temperature and the microscopy and culture test were carried out on the same day. For participants with tinea lesions, we followed the CLSI-m54a methods described by the Clinical and Laboratory Standards Institute (CLSI). The skin scrapings were inoculated onto pre-labelled Mycosel agar with a sample ID and date [71]. The agar was incubated at room temperature, and photos were taken every week for a period of eight weeks. A photo of the lesion was taken upon obtaining consent from the participant (Fig 3).

A dual-language pre-tested interview-based questionnaire in English and Bahasa Melayu was conducted face-to-face in Bahasa Melayu. For young children, the interview was carried out with the assistance of their guardians. The information collected included basic demographic data, socioeconomic status, education level, lifestyle factors, source of water supply, infrastructure availability, personal hygiene practices, as well as history of medication and skin infections. Height and weight were measured and recorded as well. The BMI was calculated and categorized into four categories (obese, overweight, normal and underweight) based on the standard set up by WHO [72] and the Ministry of Health, Malaysia.



Fig 4. Photo of skin infections via microscopy and PCR. (A) Mites in KOH processed skin sample (B) Hyphal elements in KOH processed skin sample; (C) Mycosel plates showing fungal colonies with red arrows pointing at presumptive colonies of dermatophytes; (D) *T. concentricum* colony subcultured from the primary Mycosel plate; (E) Agarose gel electrophoresis of the Pan-derm PCR products (~450bp); (F) Agarose gel electrophoresis of the ITS1-4 PCR products (~650bp).

#### c Detection of scabies and tinea

The potassium hydroxide (KOH) skin test was used for rapid detection of skin infections [70]. Briefly, the collected skin scrapings were inoculated onto the surface of a clean glass slide, followed by the addition of a drop of 20% KOH solution. The slide was then incubated at 60°C for 15 minutes. The samples were covered with cover slips and observed under light microscope. The presence of hyphal elements or mites was recorded in writing and/or by taking photographs (see Fig 4A and 4B).

For the collected Mycosel agar (Fig 4C), unique fungal colonies were sub-cultured onto Sabouraud dextrose agar (Fig 4D). The DNA was then extracted using the heat treatment method [73]. Polymerase chain reaction (PCR) (panDerm-PCR and ITS PCR) was used for the rapid confirmation of dermatophytes, utilizing the dermatophyte-specific chitin synthase primer (panDerm) [74] (Fig 4E). The isolates that tested positive were further subjected to Internal Transcribed Spacer PCR to confirm their identities [75] (Fig 4F).

#### d Statistical analysis

The data entry of the collected questionnaire was done using Microsoft Excel Pro 2019. The output file was saved in the form of comma-separated files (CSV) and stored for analysis. Statistical analysis was performed using R-4.1.2. The display graphs were created using R Statistical Software (version 4.0.3).

The discrete variables and demographic data of the study population (demographic, socioeconomic, lifestyles, and personal hygiene practices) and incidence rates of skin infections were expressed as percentages and visualized using either the ggplot2 or ggpubr package [76,77]. The normality of all continuous variables was evaluated using the Shapiro-Wilk test, with normality defined as P > 0.05.

For the risk factor analysis, Pearson's chi-square test was used to determine the association between skin infections (dependent variable) with various lifestyle and demographic variables. A cross-table output was generated using the R finalfit package [78]. The significant variables were then included in a binomial logistic regression analysis, which attempts to predict the probability that an observation falls into one of two categories of a dichotomous dependent variable based on one or more independent variables (either continuous or categorical, using the finalfit package of R [78]. The output data was interpreted using adjusted odds ratios and 95% confidence intervals. The level of significance was set at P < 0.05.

# Supporting information

**S1** Table. Three main tribes and 18 subtribes of Orang Asli in Peninsular Malaysia. (DOCX)

S2 Table. Cross-table analysis (Chi-square test) of scabies. (DOCX)

**S3** Table. Output of cross-table (Chi-square test) analysis of tinea versicolor. (DOCX)

**S4** Table. Output of cross-table (Chi-square test) analysis of tinea imbricata. (DOCX)

**S5 Table. Descriptive data of the tinea imbricata patients.** (DOCX)

# Acknowledgments

The authors would like to thank the Department of Orang Asli Development (JAKOA), Ministry of Rural and Regional Development and the village chieftains (Tok Batin) for granting the permission to enter the study villages and their cooperation during the whole course of this study. We also thank all the participants for their voluntary participation and commitment in this study. Finally, special thanks are extended to the postgraduate and undergraduate volunteers for their valuable assistance during the fieldtrips.

# **Author Contributions**

Conceptualization: Yi Xian Er, Soo Ching Lee, Yvonne Ai Lian Lim.

**Data curation:** Yi Xian Er, Azdayanti Muslim, Nan Jiun Yap, Mian Zi Tee, Yvonne Ai Lian Lim.

Formal analysis: Yi Xian Er, Soo Ching Lee.

Funding acquisition: Yi Xian Er, Azdayanti Muslim, Yvonne Ai Lian Lim.

**Investigation:** Yi Xian Er, Leslie Thian Lung Than, Azdayanti Muslim, Nan Jiun Yap, Mian Zi Tee, Nurmanisha Abdull-Majid, Shezryna Shahrizal.

Methodology: Yi Xian Er, Azdayanti Muslim, Yvonne Ai Lian Lim.

- **Project administration:** Yi Xian Er, Azdayanti Muslim, Nan Jiun Yap, Mian Zi Tee, Nurmanisha Abdull-Majid, Shezryna Shahrizal, Yvonne Ai Lian Lim.
- **Resources:** Yi Xian Er, Leslie Thian Lung Than, Azdayanti Muslim, Mian Zi Tee, Yvonne Ai Lian Lim.

Software: Yi Xian Er.

- Supervision: Leslie Thian Lung Than, Azdayanti Muslim, Yvonne Ai Lian Lim.
- Validation: Leslie Thian Lung Than, Azdayanti Muslim, Soo Ching Lee, Yvonne Ai Lian Lim.
- Visualization: Yi Xian Er, Nurmanisha Abdull-Majid, Yvonne Ai Lian Lim.
- Writing original draft: Yi Xian Er, Yvonne Ai Lian Lim.
- Writing review & editing: Yi Xian Er, Leslie Thian Lung Than, Azdayanti Muslim, Nan Jiun Yap, Mian Zi Tee, Nurmanisha Abdull-Majid, Soo Ching Lee, Shezryna Shahrizal, Yvonne Ai Lian Lim.

#### References

- Yakupu A., Aimaier R., Yuan B., Chen B., Cheng J., Zhao Y., et al. The burden of skin and subcutaneous diseases: findings from the global burden of disease study 2019. Front Public Health. 2023; 11:1145513. https://doi.org/10.3389/fpubh.2023.1145513 PMID: 37139398
- Gramp P., Gramp D. Scabies in remote Aboriginal and Torres Strait Islander populations in Australia: A narrative review. PLoS Negl Trop Dis. 2021; 15(9):e0009751. <u>https://doi.org/10.1371/journal.pntd.</u> 0009751 PMID: 34591843
- 3. Gerry G. Gambaran Distribusi Penderita Penyakit Tinea Imbrikata Berdasarkan Keadaan Sosiodemografi dan Klinis di Desa Teluk Pongkal Kecamatan Sokan Kabupaten Melawi Provinsi Kalimantan Barat Tahun 2010: Tanjungpura University; 2013.
- Polunin I. Tinea imbricata in Malaya. British Journal of Dermatology. 1952; 64(10):378–84. <u>https://doi.org/10.1111/j.1365-2133.1952.tb15801.x</u> PMID: 12987563
- Plato C.C., MacLennan R. The dermatoglyphics of the Maprik sub-district of the Sepic district of New Guinea. Zeitschrift f
  ür Morphologie und Anthropologie. 1975:208–16. PMID: 1136546
- Amare H.H., Lindtjorn B. Risk factors for scabies, tungiasis, and tinea infections among schoolchildren in southern Ethiopia: A cross-sectional Bayesian multilevel model. PLoS Negl Trop Dis. 2021; 15(10): e0009816. https://doi.org/10.1371/journal.pntd.0009816 PMID: 34613968
- 7. Hotez P.J. Aboriginal Populations and Their Neglected Tropical Diseases. Plos Neglected Tropical Diseases. 2014; 8(1):4. https://doi.org/10.1371/journal.pntd.0002286 PMID: 24498442
- White T.C., Findley K., Thomas L., Dawson J., Scheynius A., Boekhout T., et al. Fungi on the Skin: Dermatophytes and Malassezia. Cold Spring Harbors perspectives in medicine. 2014; 4(8):a019802. https://doi.org/10.1101/cshperspect.a019802 PMID: 25085959
- Szlávicz E., Németh C., Szepes É., Gyömörei C., Gyulai R., Lengyel Z. Congenital ichthyosis associated with Trichophyton rubrum tinea, imitating drug hypersensitivity reaction. Med Mycol Case Rep. 2020; 29:15–7. https://doi.org/10.1016/j.mmcr.2020.05.004 PMID: 32547912
- Al-Hasan M., Fitzgerald S.M., Saoudian M., Krishnaswamy G. Dermatology for the practicing allergist: Tinea pedis and its complications. Clin Mol Allergy. 2004; 2(1):5. <u>https://doi.org/10.1186/1476-7961-2-5</u> PMID: 15050029
- Micali G., Lacarrubba F., Verzì A.E., Chosidow O., Schwartz R.A. Scabies: Advances in Noninvasive Diagnosis. Plos Neglected Tropical Diseases. 2016; 10(6):13. https://doi.org/10.1371/journal.pntd. 0004691 PMID: 27311065
- Diaz J.H. Mite-Transmitted Dermatoses and Infectious Diseases in Returning Travelers Journal of Travel Medicine. 2010; 17(1):11.
- Johnston G., Sladden M. Scabies: diagnosis and treatment. Bmj. 2005; 331(7517):619–22. <u>https://doi.org/10.1136/bmj.331.7517.619</u> PMID: 16166133
- Romani L., Whitfeld M.J., Koroivueta J., Kama M., Wand H., Tikoduadua L., et al. The Epidemiology of Scabies and Impetigo in Relation to Demographic and Residential Characteristics: Baseline Findings from the Skin Health Intervention Fiji Trial. Am J Trop Med Hyg. 2017; 97(3):845–50. https://doi.org/10. 4269/ajtmh.16-0753 PMID: 28722612

- Walker S.L., Lebas E., De Sario V., Deyasso Z., Doni S.N., Marks M., et al. The prevalence and association with health-related quality of life of tungiasis and scabies in schoolchildren in southern Ethiopia. PLoS Negl Trop Dis. 2017; 11(8):e0005808. https://doi.org/10.1371/journal.pntd.0005808 PMID: 28771469
- JAKOA. Carta Taburan Etnik Orang Asli Mengikut Negeri: JAKOA; 2024 [Available from: https://www. jakoa.gov.my/orang-asli/carta-taburan-etnik-orang-asli-mengikut-negeri/.
- Moktar N., Noorhayati M., Fariza N., Rohani A. H., Zainal Abidin A. Health status of Orang Asli (Aborigine) community in Pos Piah, Sungai Siput, Perak, Malaysia. Southeast Asian Journal of Tropical Medicine and Public Health. 1998; 29(1):4.
- Masron T., Masami F., Ismail N. Orang Asli in Peninsular Malaysia: Population, Spatial Distribution and socio-economic condition. Journal of Ritsumeikan Social Sciences and Humanities. 2013; 6(7).
- 19. Jamiran, M.N.S., Seow, T.W., editors. Socio-economic development of the Orang Asli in Malaysia2013.
- Sharvil D. Tinea imbricata in a European: double infection with Trichophyton concentricum and Trichophyton rubrum. British Journal of dermatology. 1952; 64(10):5.
- Normaznah Y., Saniah K., Nazma M., Mak J.W., Krishnasamy M., Hakim S.L. Seroprevalence of Sarcoptes scabiei var canis antibodies among aborigines in peninsular Malaysia. Southeast Asian J Trop Med Public Health. 1996; 27(1):53–6. PMID: <u>9031401</u>
- Ahmad R. Malaysia's oldest nomads struggle to find a home London: Thomson Reuters Corporation; 2009 [updated 2009. Available from: https://www.reuters.com/article/us-malaysia-indigenousidUSTRE5A508U20091106/.
- 23. Lah A.T. Orang Asli Batek alami kulit bersisik. Utusan Malaysia. 2018.
- Leung A.K., Leung K., Lam J.M. Tinea imbricata. Journal of pediatrics. 2018; 200:285. <u>https://doi.org/10.1016/j.jpeds.2018.04.012</u> PMID: 29747932
- 25. Zahari B.J. Batek tribe told to seek immediate treatment for fungal. New Straits Times. 2019 31/3/2019.
- Roslan S.R., Hadi A.A. A child with unique skin pattern: A case report of Tinea imbricata. Medical Journal of Malaysia. 2022; 77(1):784–5. PMID: 35087009
- Seow R. Response to Minimum Wage Shock in Malaysia: SMEs' Channel of Adjustment. Journal of Economics and Sustainability (JES). 2023; 5:1–19.
- Ashley. New minimum wage gazetted Kuala Lumpur: Star Media Group Berhad; 2020 [updated Jan 14th, 2020. Available from: https://www.thestar.com.my/news/nation/2020/01/14/new-minimum-wagegazetted.
- 29. Roslan M.R. Kadar garis kemiskinan di negara ini disemak. Kosmo! 2022 March 7th, 2022.
- Shah N.M., Rus R., Mustapha R., Hussain M., Wahab N. The Orang Asli profile in peninsular Malaysia: background & challenges. International Journal of Academic Research in Business and Social Sciences. 2018; 8(7):1157–64.
- May P.J., Tong S.Y.C., Steer A.C., Currie B.J., Andrews R.M., Carapetis J.R., Bowen A.C. Treatment, prevention and public health management of impetigo, scabies, crusted scabies and fungal skin infections in endemic populations: a systematic review. Trop Med Int Health. 2019; 24(3):280–93. https://doi. org/10.1111/tmi.13198 PMID: 30582783
- Andrews R.M., McCarthy J., Carapetis J.R., Currie B.J. Skin disorders, including pyoderma, scabies, and tinea infections. Pediatr Clin North Am. 2009; 56(6):1421–40. https://doi.org/10.1016/j.pcl.2009.09. 002 PMID: 19962029
- Hylwa S.A., Loss L., Grassi M. Crusted scabies and tinea corporis after treatment of presumed bullous pemphigoid. Cutis. 2013; 92(4):193–8. PMID: 24195092
- Muslim A., Mohd Sofian S., Shaari S.A., Hoh B-P., Lim YA-L. Prevalence, intensity and associated risk factors of soil transmitted helminth infections: A comparison between Negritos (indigenous) in inland jungle and those in resettlement at town peripheries. PLOS Neglected Tropical Diseases. 2019; 13(4): e0007331. https://doi.org/10.1371/journal.pntd.0007331 PMID: 31009476
- Lin K.G. Resettlement and nutritional implications: the case of Orang Asli in regroupment schemes. Pertanika J Soc Sci Humanit. 1994; 2(2):123–32.
- Lee H., Yew V.W.C., Azima A., Sia M.K., Chan G. Perubahan sosioekonomi komuniti orang Asli Jakun akibat Rancangan Penempatan Semula: Satu kajian kes di RPS Runchang, Pahang. Malaysian Journal of Society and Space. 2018;14.
- Ebuh G.U., Salisu A., Oboh V., Usman N. A test for the contributions of urban and rural inflation to inflation persistence in Nigeria. Macroeconomics and Finance in Emerging Market Economies. 2023; 16 (2):222–46.
- 38. Xuechun Z. The urban-rural differences of inflation in China. J Finance Res. 2010; 10(004).
- 39. Zakarya S.Z. Putus bekalan makanan, orang asli pindah ke hutan. Sinarharian. 2020.

- Petrou S., Kupek E. Poverty and childhood undernutrition in developing countries: a multi-national cohort study. Social science & medicine. 2010; 71(7):1366–73. <u>https://doi.org/10.1016/j.socscimed.</u> 2010.06.038 PMID: 20691525
- Popkin B.M., Corvalan C., Grummer-Strawn L.M. Dynamics of the double burden of malnutrition and the changing nutrition reality. The Lancet. 2020; 395(10217):65–74. https://doi.org/10.1016/S0140-6736(19)32497-3 PMID: 31852602
- May P.J., Tong S.Y., Steer A.C., Currie B.J., Andrews R.M., Carapetis J.R., Bowen A.C. Treatment, prevention and public health management of impetigo, scabies, crusted scabies and fungal skin infections in endemic populations: a systematic review. Tropical Medicine & International Health. 2019; 24 (3):280–93.
- Ravine D., Turner K.J., Alpers M.P. Genetic inheritance of susceptibility to tinea imbricata. Journal of Medical Genetics. 1980; 17(5):7. https://doi.org/10.1136/jmg.17.5.342 PMID: 7218274
- Budastutik S.I., Suwarn M.L. Trichophyton concentricum as a cause of tinea imbricata. Journal of Pakistan Association of Dermatologists. 2019; 29(2):165–9.
- Tanenbaum L., Anderson C., Rosenberg M.J., Akers W. 1% sulconazole cream v 2% miconazole cream in the treatment of tinea versicolor. A double-blind, multicenter study. Arch Dermatol. 1984; 120 (2):216–9. PMID: 6364994
- Hay R.J. 77—Superficial Mycoses. In: Magill AJ, Hill DR, Solomon T, Ryan ET, editors. Hunter's Tropical Medicine and Emerging Infectious Disease (Ninth Edition). London: W.B. Saunders; 2013. p. 610–5.
- Newland J.G., Abdel-Rahman S.M. Update on terbinafine with a focus on dermatophytoses. Clin Cosmet Investig Dermatol. 2009; 2:49–63. https://doi.org/10.2147/ccid.s3690 PMID: 21436968
- Wu Z., Chen L. Optimizing the spatial arrangement of trees in residential neighborhoods for better cooling effects: Integrating modeling with in-situ measurements. Landscape and Urban Planning. 2017; 167:463–72.
- Mendez-Tovar L.J. Pathogenesis of dermatophytosis and tinea versicolor. Clinics in dermatology. 2010; 28(2):185–9. https://doi.org/10.1016/j.clindermatol.2009.12.015 PMID: 20347661
- 50. Beswick J. Exploring Eye: The Batek tribe of Malaysia and their architecture. The Architectural Review. 2010 September 22nd, 2010.
- Kaur G.A., Nadeswary K. Field trials on the management of scabies in Jengka Triangle, Pahang. Med J Malaysia. 1980; 35(1):14–21. PMID: 6166839
- Yap FB-B., Elena E.M.T., Pubalan M. Prevalence of Scabies and Head Lice Among Students of Secondary Boarding Schools in Kuching, Sarawak, Malaysia. The Pediatric Infectious Disease Journal. 2010; 29(7):682–3. https://doi.org/10.1097/INF.0b013e3181df61dd PMID: 20574407
- Arega B., Diro E., Zewude T., Getahun T., Agunie A., Owiti P., et al. High levels of scabies and malnutrition amongst orphans referred to a hospital in Addis Ababa, Ethiopia. The Journals of Infection in Developing Countries. 2020; 14(6):5. https://doi.org/10.3855/jidc.11712 PMID: 32614796
- 54. Tan H-H. Superficial fungal infections seen at the National Skin Centre, Singapore. Nippon Ishinkin Gakkai Zasshi. 2005; 46(2):77–80. https://doi.org/10.3314/jjmm.46.77 PMID: 15864251
- Urban K., Chu S., Giesey R.L., Mehrmal S., Uppal P. Burden of skin disease and associated socioeconomic status in Asia: A cross-sectional analysis from the Global Burden of Disease Study 1990–2017. JAAD International. 2021; 2:11.
- Adesiji Y.O., Omolade B.F., Aderibigbe I.A., Ogungbe O.V., Adefioye O.A., Adedokun S.A., Adekanle M.A., and Ojedele R.O. (2019). Prevalence of Tinea Capitis among Children in Osogbo, Nigeria, and the Associated Risk Factors. diseases 7, 10. https://doi.org/10.3390/diseases7010013 PMID: 30691234
- La Vincente S., Kearns T., Connors C., Cameron S., Carapetis J., Andrews R. Community Management of Endemic Scabies in Remote Aboriginal Communities of Northern Australia: Low Treatment Uptake and High Ongoing Acquisition. PLOS Neglected Tropical Diseases. 2009; 3(5):e444. https://doi.org/10.1371/journal.pntd.0000444 PMID: 19478832
- Rao G., Kuruvilla M., Kumar P., Vinod V. Clinico-epidermiological studies on tinea versicolor. Indian Journal of Dermatology, Venereology and Leprology. 2002; 68(4):2. PMID: 17656938
- He S., W-D D., Yang S., S-M, Z., W, L., Wang J., et al. The genetic epidemiology of tinea versicolor in China. Mycoses. 2008; 51(1):8. https://doi.org/10.1111/j.1439-0507.2007.01437.x PMID: 18076596
- Heidrich D., Daboit T.C., Stopiglia C.D., Magagnin C.M., Vetoratto G., Amaro T.G., Scroferneker M.L. SIXTEEN YEARS OF PITYRIASIS VERSICOLOR IN METROPOLITAN AREA OF PORTO ALEGRE, SOUTHERN BRAZIL. Rev Inst Med Trop Sao Paulo. 2015; 57(4):277–80. https://doi.org/10.1590/ S0036-46652015000400001 PMID: 26422149

- Makrantonaki E., Ganceviciene R., Zouboulis C. An update on the role of the sebaceous gland in the pathogenesis of acne. Dermatoendocrinol. 2011; 3(1):41–9. https://doi.org/10.4161/derm.3.1.13900 PMID: 21519409
- Eijsvogels T.M., Veltmeijer M.T., Schreuder T.H., Poelkens F., Thijssen D.H., Hopman M.T. The impact of obesity on physiological responses during prolonged exercise. International journal of obesity. 2011; 35(11):1404–12. https://doi.org/10.1038/ijo.2010.277 PMID: 21266953
- Ng K.P., Soo-Hoo T.S., Na S.L., Ang L.S. Dermatophytes isolated from patients in University Hospital, Kuala Lumpur, Malaysia. Mycopathologia. 2002; 155(4):203–6. https://doi.org/10.1023/ a:1021194025765 PMID: 12650596
- Sahu R.P., Petrache I., Van Demark M.J., Rashid B.M., Ocana J.A., Tang Y., et al. Cigarette Smoke Exposure Inhibits Contact Hypersensitivity via the Generation of Platelet-Activating Factor Agonists. The Journal of Immunology. 2013; 190(5):2447–54. https://doi.org/10.4049/jimmunol.1202699 PMID: 23355733
- 65. Yazdanparast T., Hassanzadeh H., Nasrollahi SA., Seyedmehdi SM., Jamaati H., Naimian A., et al. Cigarettes Smoking and Skin: A Comparison Study of the Biophysical Properties of Skin in Smokers and Non-Smokers. Tanaffos. 2019; 18(2):163–8. PMID: 32440305
- Jamalunlaili A., Sayuti N., Arshad A., Embong M. Living Conditions in Orang Asli Resettlement Project (PROSDET) of Pantos, Pahang, Malaysia. Procedia—Social and Behavioral Sciences. 2016; 222:143– 50.
- 67. Esri, Michael Bauer Research GmbH, Department of Statistics Malaysia, cartographers. Malaysia Negeri Boudaries Provides a 2022 Boundary with a total population count2022.
- 68. Cohen J. Statistical Power Analysis for the Behavioral Sciences. New York: Routledge; 1988.
- Champely, S. Package 'pwr' 2022 [updated October 14, 2022. Available from: https://cran.r-project.org/ web/packages/pwr/pwr.pdf.
- 70. Ponka D. Microscopic potassium hydroxide preparation. Canadian Family Physician. 2014; 60(1):1. PMID: 24452564
- 71. Wengenack N. CLSI M54A: Principles and procedures for detection of fungi in clinical specimens— Direct examination and culture. CLSI (1st edition)2012.
- 72. Consultation W. Obesity: preventing and managing the global epidemic. World Health Organization technical report series. 2000; 894:1–253.
- 73. Dashti A., Jadaon M., Abdulsamad A., Dashti H. Heat Treatment of Bacteria: A Simple Method of DNA Extraction for Molecular Techniques. Kuwait Medical Journal. 2009; 41.
- 74. Garg J., Tilak R., Singh S., Gulati A.K., Garg A., Prakash P., Nath G. Evaluation of pan-dermatophyte nested PCR in diagnosis of onychomycosis. J Clin Microbiol. 2007; 45(10):3443–5. https://doi.org/10. 1128/JCM.02367-06 PMID: 17699656
- Martin K.J., Rygiewicz P.T. Fungal-specific PCR primers developed for analysis of the ITS region of environmental DNA extracts. BMC Microbiology. 2005; 5(1):28. https://doi.org/10.1186/1471-2180-5-28 PMID: 15904497
- 76. Kassambara A. ggpubr: 'ggplot2' Based Publication Ready Plots (R package version 0.4.0) 2021 [Available from: https://rpkgs.datanovia.com/ggpubr/.
- 77. Wickham H., Chang W., Henry L., Pedersen T. L., Takahashi K., Wilke C., et al. ggplot2: Elegant Graphics for Data Analysis. New York: Springer-Verlag New York; 2016.
- 78. Harrison E., Drake T., Ots R. finalfit: Quickly Create Elegant Regression Results Tables and Plots when Modelling 2023 [Available from: https://cran.r-project.org/web/packages/finalfit/index.html.